Author's response to reviews

Title: Evidence based guidelines for complex regional pain syndrome type 1.

Authors:

    Roberto SGM Perez (rsgm.perez@vumc.nl)
    Paul E Zollinger (pe.zollinger@tiscali.nl)
    Pieter U Dijkstra (p.u.dijkstra@rev.umcg.nl)
    Ilona L Thomassen-Hilgersom (IlonaThomassen@chello.nl)
    Wouter WA Zuurmond (wwa.zuurmond@vumc.nl)
    Kitty (C) JGM Rosenbrand (k.rosenbrand@cbo.nl)
    Jan HB Geertzen (j.h.b.geertzen@rev.umcg.nl)

Version: 5 Date: 29 March 2010

Author's response to reviews:

Dear Editor-In-Chieff,

We thank the editorial staff for accepting our manuscript entitled “Evidence based guidelines for complex regional pain syndrome type 1” (MS: 4828390682892911). Please find point by point description of the changes made in this version of the manuscript.

Yours sincerely,

On behalf of all co-authors,

Roberto Perez

Major revisions:
1. The description of authors' names has been changed as requested.
2. A conclusion section was added as requested.
3. A statement was provided to the authors contributions that all authors read and approved the final manuscript.
4. Table 1 has been merged into one complete table.
5. Table 2 has been changed into an additional file.
6. References to additional files have been changed in the text as requested.
7. A section “Additional files” has been added as requested.

Minor revisions:
1. The title has been placed in sentence case.
2. Email addresses have been provided for all authors on teh title page.
3. Highlighting has been removed.
4. Introductions was renamed “Background”
5. Table and figure placement in the text was removed.
6. The manuscript was checked.